Author's response to reviews

Title: Development of functional gastrointestinal disorders after Giardia lamblia infection

Authors:

Kurt K Hanevik (mfakh@ikb.uib.no)
Vernes V Dizdar (vedi@helse-bergen.no)
Nina N Langeland (nila@helse-bergen.no)
Trygve T Hausken (tryh@helse-bergen.no)

Version: 4 Date: 20 April 2009

Author's response to reviews:

Some minor corrections have been made. A few commas has been added and a "the" has been taken out in the abstract, and an "%" added.

A wrong number in the result section of the abstract has been corrected to "57.7%" consistent with the number in table 1.

"influenced" on page 5, second paragraph has been changed to "worsened" to clarify the quality of the influence.

"A strength of this study ..." starts a new paragraph and this has been corrected.

No asterix for the corresponding author can bee seen behind the name. Is this correct?

With kind regards,

Kurt Hanevik